Author's response to reviews

Title: The effect of five years versus two years of specialised assertive intervention for first episode psychosis - OPUS II: study protocol for a randomized controlled trial

Authors:

Marianne Melau (marianne.melau@regionh.dk)
Pia Jeppesen (pia.jep@dadlnet.dk)
Anne Thorup (atv@dadlnet.dk)
Mette Bertelsen (mbe_76@yahoo.dk)
Lone Petersen (lonepetersen@comxnet.dk)
Christian Gluud (cgluud@ctu.rh.dk)
Gertrud Krarup (gerkrar@rm.dk)
Merete Nordentoft (merete.nordentoft@dadlnet.dk)

Version: 6 Date: 3 March 2011

Author's response to reviews:

I add the following sentence in the abstract's background section:

"The major objective of the OPUS II trial is to evaluate the effects of five years of OPUS treatment versus two years of OPUS treatment."

In the main manuscript I changes "positive symptoms" to "psychotic symptoms" where it was about the psychopathology.